Author's response to reviews

Title: Synthetic Fiber from a Teddy Bear Causing Keratitis and Conjunctival Granuloma: Case report

Authors:

Mohammed K Farooq (kashfar@gmail.com)
Jan U Prause (jup@sund.ku.dk)
Steffen Heegard (sthe@sund.ku.dk)

Version: 5 Date: 19 June 2011

Author's response to reviews:

To the Chief Editor of BMC ophthalmology, 19.06.2011

We have made a few typographical changes in the manuscript. The details are given below.

1. There have been added "of the teddy bear" in the abstract and in the results section: A) Hairs from the face of the teddy bear were morphologically and microscopically identical with the fibers causing the severe corneal ulceration in the two-year-old girl.

2. The authors contribution section has been change to the following: . SH and JUP performed the microscopical examination of the cornea and the conjunctival granuloma and assisted in the final preparation and submission of the manuscript.

With kind regards,

Prof. Steffen Heegaard
Eye Pathology Section, University of Copenhagen, Denmark.